Author's response to reviews

Title: Primary hydatid cyst of the adrenal gland resected laparoscopically

Authors:

Gianlorenzo Dionigi (gianlorenzo.dionigi@uninsubria.it)
Gianpaolo Carrafiello (gcarraf@tin.it)
Chiara Recaldini (chiarec@libero.it)
Fausto Sessa (fausto.sessa@uninsubria.it)
Luigi Boni (luigi.boni@uninsubria.it)
Francesca Rovera (francesca.rovera@uninsubria.it)
Renzo Dionigi (renzo.dionigi@uninsubria.it)

Version: 4 Date: 18 June 2007

Author's response to reviews:

Dear Reviewer,

Thank You for Your general comments on this manuscript “PRIMARY HYDATID CYST OF THE ADRENAL GLAND: DESCRIPTION OF A CASE AND REVIEW OF THE LITERATURE”.

Thank you very much for Your collaboration and interest for reviewing our paper. Following the reviewers suggestions, a major revision of the English language was made, key point and additional information were added, showing the key aspects of the manuscript and the salient issues regarding the primary hydatid cyst of the adrenal gland treatment.

Thank You very much for Your collaboration

Kindest Regards,

Gianlorenzo Dionigi